EO-1 Weekly Status Week of February 12 – February 18, 2009

Day of Year 043 - 049

Mission Day 3013 - 3019

Earth Observing-One (EO-1) General

Work has started on preparing the Senior Review Proposal for EO-1 operations for FY10-11.  It is due at NASA Headquarters on March 24 and an award decision will be made July-August 2009.

There were 67 Data Collection Events (DCEs) scheduled this week. 

INSTRUMENTS

All instruments operated nominally this week 

ALI Instrument dark current investigation test for the LDCM project:

· Limited imaging (approximately one image every two days) being conducted from February 16 to March 4.

· This experiment is to better characterize drifts in the ALI dark current calibration.

EO-1 Spacecraft Subsystems

Command and Data Handling (C&DH)

Continued to experience problems playing back engineering data from the Solid-State-Recorders. Engineering data is being received during real-time contacts and for all science imaging events. 
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In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

SensorWeb & Virtual Observatory Demonstrations

A teleconference between the Flood SensorWeb collaborators was held on February 17.  The following items of interest were discussed:

1..  The Canadian Space Agency has setup an internal project that authorizes the Flood SensorWeb to task RADARSAT-2 for data delivery for future floods as a prototype concept without having to activate the International Charter.

2.  RADARSAT-2 images of Australian flooding were taken on February 14th and 17th. The data from these two images will be sent to the National Space Agency of Ukraine (NSAU), the Australian Department of Climate Change, and Geoscience Australia where collaborators will construct flood maps.

3.  The Dartmouth Flood Observatory has produced flood maps for the flooded areas in Mozambique, Madagascar, and Normanton, Australia.  They were asked to setup the capability to deliver these kinds of flood maps automatically on demand.

4.  An effort is being made to schedule another attempt to obtain EO-1 images of the Mozambique and Normanton flood areas to complement FORMOSAT-2 and RADARSAT-2 images obtained earlier.  Previous attempts two weeks ago for obtaining EO-1 imagery of these regions failed due to it being too cloudy or not obtained due to a Hyperion anomaly.

5.  Dan Mandl has communicated with Guy Seguin of the Canadian Space Agency concerning his invitation to present Flood SensorWeb status at the CEOS SIT meeting on March 3.  Dan has sent an abstract about his talk that will be presented at the Disaster Management session.

7.  An attempt will be made to overlay the Normanton flood image data on a Google map to show flooded roads.

8.  The International Red Cross representative, Frederic Zanetta,  recommended that we should read their reports on the flooding in Australia to see if we could find related reports to the images which we acquired and processed.
9.  Stu Frye is preparing input to a memorandum to be sent from the Group on Earth Observations in Geneva to the International Disaster Charter outlining plans for expanding access to GEO member nations for International Charter activation and to accept and act on Flood SensorWeb triggers that will target future events 1-3 days out instead of waiting on a disaster to occur before scheduling observations.

10. EO-1 obtained some impressive images of the wildfires in Bright, Australia that are displayed on the NASA Earth Observatory image of the day website.  One of the images is shown below in figure 1.
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Figure 1 This EO-1 image of the Bright fires in Australia taken February 10, 2009 was created using bands 10-7-5.  Thus both active fires and burn scars can be identified.

Formosat-2 was also targeted to take the same area as part of a fire SensorWeb data run with images forthcoming.

There was a Mission Science Office meeting held on February 17.  The meeting was chaired by Elizabeth Middleton and was attended by Petya Campbell, Lawrence Ong, Steve Ungar, Dan Mandl; Stu Frye, Kim Hall, Paul Brandinger, and Joe Young.  The meeting focused entirely on defining activities, budget and schedules for preparing the EO-1 Senior Review proposal. 

Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week. 

EO-1 MISSION OPERATIONS CENTER

Real-Time

Spacecraft state of health is currently nominal.

The EO-1 team at GSFC and Wallops Flight Facility has setup new firewall rules to accommodate the Air Force Phased Array Demonstration test.  A telecon was held with Air Force personnel at Schriever AFB in Colorado Springs, CO and Onizuka Air Force Station in Sunnyvale, CA to finalize the uplink and downlink configurations and transfer rates.  Testing should begin in March where EO-1 will transmit S-band telemetry to and receive commands from the phased array at Schriever AFB.
EO-1 MOC failed to receive spacecraft dumps on February 17:

· FOT switched operations to backup T&C workstation

· FOT successfully received dumps through manual commanding on February 18

· Thereafter, MOC successfully used automation on backup T&C workstation to receive spacecraft dumps

· Investigation of cause is currently underway 

Flight Dynamics

PF2 tracking data issues:

· PF2 is continuing to experience problems with tracking data for EO-1

· FOT has contacted PF2 to assist with troubleshooting.

· Large Doppler biases required for PF2 data

· Even with large Doppler biases, observed range rates for PF2 do not agree well enough with other stations to include in orbit determination calculations

MLT Control:

· FOT will continue to perform MLT control maneuvers to maintain the MLT at its current value.

· Next set of maneuvers are being planned for February 24 and February 26.

· MLT will simply be maintained at 10:00:00 throughout rest of mission

System Administration

Firewall Status:

· Primary firewall is back online with full functionality.

· Firewall updated to give JPL access to multiple machines in EO-1 MOC.

ASPEN

· ASPEN boxes have been placed on network after successful system scans

ITPS

· Some medium-risk entries found in scan of ITPS box

· Results of the scan are under investigation by the security group

· ITPS box will not be put on network until issue is resolved

Bandwidth

· Slowdown between EO-1 firewall and the IP Networks Operations Center (IPNOC) routers has been resolved

· Code updated to utilize correct settings for firewall network cards when rebooted

Development

· Axle rebuilt with Red Hat Enterprise 4 and ASIST 9.7.j

· Still troubleshooting installation of Firefox

Security

· Meeting conducted to determine actions necessary before security audit (week of March 1)

· Research of relevant documents was conducted by FOT to determine how recent Sensitive But Unclassified (SBU) policies affect EO-1 operations

· Meeting being planned to further discuss issue

GROUND AND SPACE NETWORK

Station Downtimes

PF1 has been experiencing problems with their universal matrix switch and has been oscillating between red and yellow status since January 22:

· EO-1 schedule is utilizing PF2 instead of PF1 until PF1 appears stable
Operational Discrepancies 

No major problems to report.

UPCOMING EVENTS

MLT control maneuvers:

· FOT will conduct two inclination burns to continue to control the MLT.

· First maneuver will be on February 24, 2009 at 15:21:57 UTC.

· Second maneuver will be on February 26, 2009 at 15:00:21 UTCs.

· Both maneuvers will be 350 seconds in duration.

Imagery Status

Scenes and Engineering Cals planned for week of February 12 – February 18, 2009             67

Total scenes and engineering calibrations planned for entire mission – approximately 44,202

Total Scenes:  ALI scenes in the Level 0 archive              40,096 (as of February 18, 2009)

                         Hyperion scenes in the Level 0 archive    39,848
Publications and Presentations Status 

404 publications 

284 external presentations

53 articles and press releases
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